Author's response to reviews

Title: The economic impact of diabetes through lost labour force participation on individuals and government: evidence from a microsimulation model

Authors:

Deborah Schofield (deborah.schofield@ctc.usyd.edu.au)
Rupendra N Shrestha (rupendra.shrestha@ctc.usyd.edu.au)
Megan E Passey (Megan.Passey@ucrh.edu.au)
Lennert Veerman (l.veerman@uq.edu.au)
Emily J Callander (emily.callander@ctc.usyd.edu.au)
Michelle Cunich (michelle.cunich@ctc.usyd.edu.au)
Simon Kelly (Simon.Kelly@natsem.canberra.edu.au)
Robert Tanton (Robert.Tanton@natsem.canberra.edu.au)

Version: 2 Date: 17 September 2013

Author's response to reviews: see over